Author's response to reviews

Title: Self-management of type 2 diabetes mellitus: a qualitative investigation from the perspective of participants in a nurse-led, shared-care programme in the Netherlands

Authors:

Albine Moser (a.moser@zw.unimaas.nl)
Harry van der Bruggen (h.vanderbruggen@zw.unimaas.nl)
Guy Widdershoven (g.widdershoven@zw.unimaas.nl)
Cor Spreeuwenberg (c.spreeuwenberg@zw.unimaas.nl)

Version: 4 Date: 25 February 2008

Author's response to reviews: see over
Cover letter

MS: 1140900505149763
Self-management of type 2 diabetes mellitus: a qualitative investigation from the perspective of participants in a nurse-led, shared-care programme in the Netherlands Albine Moser, Harry van der Bruggen, Guy Widdershoven and Cor Spreeuwenberg

25 February 2008

Dear Dr. Browning,

Thank you for your mail.

I would like to address the following editorial comment:
However before acceptance, we strongly encourage you to include an Acknowledgements section between the Authors? contributions section and Reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements. Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

I have addressed this editorial comment accordingly, see page 19-20, section: Acknowledgments

I hope that I revised the manuscript according to the standard required for final acceptance. If you have any questions please do not hesitate and contact me.

With kind regards,

Albine Moser
On behalf of the authors